Author's response to reviews

Title: Resting heart rate as predictor of metabolic dysfunctions in obese children and adolescents

Authors:

Ismael F Freitas Junior (ismael@fct.unesp.br)
Paula A Monteiro (paulinha_1003@hotmail.com)
Loreana S Silveira (loreana_loly@hotmail.com)
Suziani U Cayres (suziungari@yahoo.com.br)
Barbara M Antunes (bah_tunes@hotmail.com)
Karolynne N Bastos (karolyneoley@yahoo.com.br)
Jamile S Codogno (jamile_sc@yahoo.com.br)
João Paulo J Sabino (jpsabino@rfi.fmep.usp.br)
Romulo A Fernandes (romulo_ef@yahoo.com.br)

Version: 4 Date: 11 January 2012

Author's response to reviews: see over
Major revisions

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is "First name Initial Last name" (e.g. Joe F Bloggs)

As solicited, the authors’ names were changed.

References: All references should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers.

As solicited, superscript numbers were excluded.

Structure: Please provide a Conclusions section, or rename the Discussion section to "Discussion and Conclusions"

As solicited, a “Conclusions Section” was created.

Authors' contributions: Please indicate in this section that all authors read and approved the final manuscript.

As solicited, this statement was inserted in the “Authors’ contributions”.

Minor revisions

Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, "The Title of My Prestigious Publication: A Scientific Study." should be changed to "The title of my prestigious publication: a scientific study"

As solicited, the title was modified.

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript, and change any coloured text to black

As solicited, highlighting and colored text were removed from the manuscript.

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements

As solicited, tables were modified.